Author's response to reviews

Title: Effectiveness of a complex intervention on reducing the prevalence of smoking among adolescents: study design of a cluster-randomised, controlled trial.

Authors:

Alfonso Leiva (aleiva@ibsalut.caib.es)
Andreu Estela (andreu.estela@ssib.es)
Maties Torrent (maties.torrent@ssib.es)
Amador Calafat (irefrea@irefrea.org)
Miquel Bennasar (miquel.bennasar@uib.es)
Aina Yáñez (ainam.yanez@ssib.es)

Version: 2 Date: 24 January 2014

Dear Editor

We have included in the study population (methods section) the consent statement as you required.

"Informed consent were obtained from all the students. Consent was also obtained from one of the adolescent's parents/guardians, according to the declaration of Helsinki"

Yours sincerely

Alfonso Leiva